

Contents lists available at ScienceDirect

Agricultural Water Management



journal homepage: www.elsevier.com/locate/agwat

Increased heat stress reduces future yields of three major crops in Pakistan's Punjab region despite intensification of irrigation.

Rike Becker^{a, b, *}, Christoph Schüth^{c, g}, Ralf Merz^d, Tasneem Khaliq^e, Muhammad Usman^f, Tim aus der Beek^g, Rohini Kumar^h, Stephan Schulz^c

^a University of Kassel, Nordbahnhofstr. 1, 37213 Witzenhausen, Germany

^b Potsdam Institute for Climate Impact Research, Telegraphenberg 31, 14473 Potsdam, Germany

^c TU Darmstadt, Schnittspahnstraße 9, 64287 Darmstadt, Germany

^d UFZ Helmholz Centre for Environmental Research, Theodor-Lieser-Straße 4, 06120 Halle, Germany

^e University of Agriculture Faisalabad, P.O. Box No 38040, Faisalabad, Pakistan

^f Martin-Luther University Halle-Wittenberg, Von-Seckendorff-Platz 4, 06120 Halle, Germany

^g IWW Water Centre, Moritzstraße 26, 45476 Mülheim an der Ruhr, Germany

h UFZ Helmholz Centre for Environmental Research, Permoserstr. 15, 04318 Leipzig, Germany

ARTICLE INFO

Handling Editor: Dr. B.E. Clothier

Keywords: Regional climate change Yield Water demand SWAT APSIM

ABSTRACT

Climate change and variability threaten the sustainability of future food production, especially in semi-arid regions where water resources are limited and irrigated agriculture is widespread. Increasing temperatures will exacerbate evaporative losses and increase plant water needs. In this regard, higher irrigation intensities have been posited as a solution to mitigate climate change impacts in these regions. Here, using the agrohydrological model SWAT and the biophysical crop model APSIM, we show that this mitigation measure is oversimplified. We find that heat stress, driven by strong temperature increases, might be the dominating factor in controlling future crop yields and plant water needs. Our analysis encompasses agricultural areas of the Lower Chenab Canal System in Punjab, Pakistan (15,000 km²), which is part of the Indus River irrigation system, the largest irrigation system in the world, covering major cotton, rice and maize cropping zones. Climate models project a strong increase in temperature over the study region of up to 1.8 °C (±0.5 °C) until the mid-century. Both models predict a decline in future crop yields for maize and rice crops, while cotton yields are less effected by rising temperatures and strongly benefit from elevated atmospheric CO₂ concentrations. For a high carbon emission scenario, the models simulate yield declines for maize of up to -10% (APSIM) and -19% (SWAT); for rice yields of up to -4% (APSIM) to -26% (SWAT), and for cotton yields of -1% (APSIM) to +11% (SWAT), until 2050, relative to the baseline scenario 1996-2005. Our modeling results further suggest that irrigation demands do not align with increasing temperature trends. Average irrigation demands increase less under higher temperatures. Overall, our study emphasizes the role of elevated heat stress, its effects on agricultural productivity as well as water demand, and its implications for climate change adaption strategies to mitigate adverse impacts in an intensively irrigated region. Plain language summary: Climate change is one of the most important challenge facing agriculture, and hence

future food security. Farmers struggle more and more to ensure a reliable food production. This is especially true for semi-arid regions where water resources are limited but at the same time urgently needed for irrigation, such as in the Lower Chenab Canal Area, in Punjab, Pakistan. The search for adaptation measures to weaken the negative impacts of climate change on agricultural systems becomes increasingly important. One option could be the intensification of irrigation. We use two models to simulate crop growth processes under various climate conditions. Our results show that heat stress would be the dominating stress for plants in the selected study region and that this heat stress might even lead to a reduction in water demand. Intensive irrigation can not necessarily help to prevent climate change-related yield losses. The models further predict that rising CO_2 can have a positive effect on crop yield by enhancing plant growth. Nevertheless, our results indicate that these positive effects could not compensate the negative impacts of heat stress under constantly rising temperatures.

* Corresponding author at: University of Kassel, Nordbahnhofstr. 1, 37213 Witzenhausen, Germany. *E-mail address:* rike.becker@uni-kassel.de (R. Becker).

https://doi.org/10.1016/j.agwat.2023.108243

Received 20 December 2022; Received in revised form 21 February 2023; Accepted 22 February 2023 Available online 27 February 2023 0378-3774/© 2023 The Authors. Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

1. Introduction

In recent years, climate change and its impact on the environment is one of the main concerns facing societies globally. Specifically, its effect on agricultural systems has become a major problem considering the alarming global developments regarding water and food security (Hanjra and Qureshi, 2010; Schewe et al., 2014). One of the latest special report of the UN Intergovernmental Panel on Climate Change (Shukla et al., 2019) predicts with high confidence that future changes in climatic conditions will exacerbate existing water and food shortages for billions of people. One of the main reasons considered responsible for the predicted food shortage is the inability to meet future agricultural water demands (Fader et al., 2016). Globally, irrigation volumes have more than doubled since the 1960s (Shukla et al., 2019) and are likely to increase further due to climate change in regions with already limited water supply (Wada et al., 2013; Wang et al., 2016).

In semi-arid and developing regions like Pakistan, agriculture is the most important economic sector, employing nearly half of the population (Qureshi, 2011). A large part of agricultural workers are small-scale farmers who are highly vulnerable to yield losses, which have become increasingly frequent in the recent decades (Oxfam, 2009). The projected increase in water scarcity due to climate change, along with the increasing demand of the fast-growing population, poses a severe threat to the national food supply and the production of economically important cash crops such as cotton, maize, and rice (Ahmad et al., 2015; Khan et al., 2016; Schewe et al., 2014).

The Indus Basin in Pakistan's Punjab province is a hot spot for climate change impacts on water availability and agricultural production, as it constitutes one of the world's largest closed irrigation areas (Mekonnen and Hoekstra, 2011). Irrigation water accounts for over 90% of the total water demand in the region (Fischer et al., 2007). Significant climate-induced changes in the upstream glacier-hydrology - the major water source for the Indus Basin - are threatening future water availability in the basin (Immerzeel et al., 2010); along with the rising temperatures which are projected to increase faster than the global average (Saeed and Athar, 2018). Under such conditions, water-related adaptation strategies, such as increased irrigation amounts, and enhanced irrigation efficiency are potential solutions to cope with these challenges. The effects of such adaptation measures have been studied for agricultural systems experiencing similar climate change pressures and have been suggested as possible actions (Elliott et al., 2014; Fader et al., 2016; Molden et al., 2010). However, the direct role of increasing temperatures in affecting crop yields is gaining importance across the globe, and could be an important factor in the Indus Basin. Temperature-induced stress on crop growth could counteract the potential of enhanced irrigation to increase productivity (Zaveri and Lobell, 2019). It is therefore imperative to understand the role of temperature stress on crop growth and resulting plant water demand in combination with water stress. Improved knowledge on the potential impacts of temperature and water stress as well as their interlinkages is important to define adequate adaptation strategies (Lesk et al., 2022). In terms of adequate water availability for crop growth, especially in semi-arid regions, previous studies highlight that there is still very limited understanding of the potentials and limits of irrigation-related climate change adaptation (Tack et al., 2017; Taraz, 2018); and that more research is needed to disentangle the effects of temperature and water stress-related climate change impacts on agricultural yields (Carter et al., 2016).

The main objective of this study is to quantify the impact of heat stress on yield and crop water demand, in the Lower Chenab Canal System, in Pakistan. We hypothesize that future temperature conditions will have a strong negative impact on crop yields in our study area. Further, we propose the counter-intuitive hypothesis: increasing temperatures will lead to a reduction in irrigation requirements driven by a strong decrease in crop growth which leads to reduced plant water needs.

To assess these hypotheses, the study elaborates on how temperature stress controls agricultural productivity and plant water requirements in an intensively irrigated agricultural system. To explore whether our counter-intuitive assumption that increasing temperatures will lead to decreasing irrigation demands, holds true, we apply two models: the hydrological SWAT model (Arnold et al., 2012) and the biophysical-crop modelling framework APSIM (Holzworth et al., 2014). While biophysical crop models are the classical models to be used to assess climate change impacts on crops, studies have shown their low performance in reproducing hydrodynamics such as soil moisture storage or plant water availability (Faria and Bowen, 2003), which are crucial processes for the estimation of water requirements and demands. Hydrological models on the other hand, show low performances in simulating plant physiological behaviour and yield estimations. In this study we therefore use a hydrological and a biophysical crop model in an ensemble framework to assess climate change impacts on crop production as well as climate change impacts on plant water demands. A hard coupling of both models is beyond the scope of this study. Yet, the profound comparative analysis of both model types is expected to allow a more detailed understanding of strengths and weaknesses of either model and will result in a more reliable assessment of changes in future yield and water demands. Uncertainties of the respective model applications are discussed in detail and give valuable insights for modellers of both modelling communities, as they reveal important limits of climate impact assessment based on process-based hydrological as well as crop models.

2. Materials and methods

2.1. Study area

The study area is part of the Lower Chenab Canal System Area (LCC) in Pakistan, which comprises about 15,000 km² of agricultural land on the floodplains between the Rivers Chenab and Ravi (Fig. 1A and B). It belongs to the Indus Basin Irrigation System (IBIS), the world's largest irrigation system, feeding more than 200 million people (Immerzeel et al., 2010). The area is characterized by small-scale and highly fragmented agricultural cropping patterns. During the dry winter season (Rabi) the dominating crop type is winter wheat while during the wetter and hot summer (Kharif) the crop pattern diversifies and mainly cotton, maize, rice, and fodder are grown on small-scale farms. Annual reference evaporation (approx. 1800 mm/a) is more than three times larger than annual precipitation (approx. 500 mm/a), resulting in a strong demand for additional irrigation. Knowing about potential negative impacts on agricultural production and defining possible adaptation strategies is therefore of paramount importance for water and food security in this region.

In this study, we focused on analysing the impacts of future climate change on summer crops, namely cotton, maize, and rice, grown between May and October. Impacts are evaluated based on changes in crop yield and relevant hydrologic and biophysical variables including evapotranspiration, irrigation demand, leaf-area growth, and biomass production. The selected crops represent high-value crops with a wide distribution in the study area (Fig. 1A). Thus, changes in their production levels will have significant economic impacts and changes in irrigation needs will have a strong impact on basin-wide water demand.

2.2. Models: SWAT and APSIM

To analyse climate change impacts on yield and water demand, we apply two models from different scientific disciplines. We consider the agro-hydrological SWAT model (Arnold et al., 2012), with high performance in simulating complex hydrological plant-soil-atmosphere interactions and the biophysical crop modelling framework APSIM (Holzworth et al., 2014), with high performance in simulating complex plant growth dynamics. The models are briefly described below:

The agro-hydrological SWAT model (Soil & Water Assessment Tool)

simulates the quantity and quality of water flow within catchments, incorporates detailed management strategies (e.g. irrigation schedules, planting schedules) and basic plant physiognomic stages, e.g. root development, leaf area development, and biomass change (Arnold et al., 2012; Gassman et al., 2014). The main underlying principle for the simulation of water fluxes is the water balance closure (Neitsch et al., 2009). SWAT accounts for spatially distributed environmental and climatic changes and it simulates their effects on individual water balance components, such as evapotranspiration, infiltration, soil moisture changes and runoff (Fig. 2). Its strengths are therefore the detection of spatially distributed changes in water availability and demand. Impacts of increasing atmospheric CO₂ concentrations are accounted for in the estimation of potential evapotranspiration, affecting (i.e., reducing) plant water demand as well as in the estimation of plant radiation use efficiency, affecting (i.e., enhancing) biomass production. To ensure correct and spatially differentiated parameterization, the model is calibrated following an automated and spatially distributed calibration approach. A detailed description of this calibration procedure can be found in Becker et al. (2019). In this study, the model was run on a daily timescale, with daily climate input data. The spatially distributed SWAT results are averaged over each respective crop growing region, resulting in one maize, one rice, and one cotton data set.

The Agricultural Production System Simulator (APSIM) is a biophysical crop modelling framework which simulates agricultural crop dynamics with respect to varying climatic and environmental conditions (Holzworth et al., 2014). It has been extensively used to assess climate change impacts on agricultural productivity (Deihimfard et al., 2018; Gaydon et al., 2021; Khaliq et al., 2019). Model performance and applications are studied in depths within the scope of the Agricultural Modelling Intercomparison and Improvement Project - AgMIP (Boote et al., 2021; Rosenzweig et al., 2014), in which APSIM was applied in the same study region of southern Punjab to assess climate change impact on crop production. Focus and strength of the APSIM framework is the plant-specific simulation of biophysical dynamics such as biomass production, yield or root growths (Fig. 2). Due to its modular approach, with individual sub-models for each crop type, it can account for plant specific reactions to climate change. For example, with individual models for cotton, rice, and maize it accounts for plant type specific carbon assimilation processes (C3 vs. C4-plants) and hence, differentiates between plant type reactions to increased atmospheric CO₂ levels.

Different to SWAT, APSIM is a plot-scale model. To allow for a model comparison between the spatially distributed SWAT model and APSIM, we calibrated the APSIM model by adopting the most important soil and management parameter configurations from the calibrated SWAT model. To rule out uncertainties which would arise due to spatial differences in soil and climate conditions over the large LCC area, we conducted a sensitivity analysis to analyse the effect of varying soil and climate parameters on APSIM simulated crop yields. This further allowed us to constrain appropriate parameters for the APSIM model. Soil parameters were furthermore verified through laboratory analysis of soil samples collected during a field campaign in the study region (Schulz et al., 2021). Details of the used parameters and underlying estimation procedure for both crop models are described in the Supplementary Information S2 and Tables S1-S3. Like the SWAT model, APSIM was run on a daily time scale, with daily climate input data.

Two central management-related assumptions were made in both models: i. water stress is minimized by including regular irrigation events, as soon as soil moisture drops below 90%; and ii. nutrient stress is minimized by assuring sufficient fertilizer application, applying 100 kg/ha of nitrogenous fertilizer at each irrigation event. Simulated trends of future yield, biomass production, evapotranspiration, LAI and water demand can therefore be ascribed to the effect of temperature increase, by ruling out the influence of irrigation and nutrient management.

The skill of both models to correctly reproduce plant water demands was validated against observed crop specific evapotranspiration (ET)



Fig. 1. (A) Lower Chenab Canal (LCC) study area and spatial distribution of cotton, maize and rice growing regions (Land-use data from Awan et al., 2016). (B) Overview of LCC study area, Pakistan, and the Indus River Basin. (C) Mean annual temperature and (D) precipitation trends of historical data (black line) and future climate projection of 9 CORDEX models (red and blue lines are ensemble means; coloured uncertainty band span between 25th and 75th percentiles). Shaded grey areas (C and D) show the reference period (1996–2005) and future periods of 2021–2030 and 2041–2050, examined in this study.



Fig. 2. Schematics of the main processes simulated by SWAT and APSIM models. Processes analysed in this study are written in bold and italic letters.

rates, for the years 2009–2010. This time period was chosen based on the availability of high resolution and crop specific ET data. To validate the model performances with respect to yield simulations annual yield estimates were validated using observed yield data provided by Agricultural Statistics of Pakistan, published by the Ministry of National Food Security & Research (MNFSR, 2021). Observed yield data could be obtained for the years 2002–2013, at the province level for the province of Punjab.

2.3. Climate data sets

Daily Climate Forecast System Reanalysis data (Saha et al., 2010) was taken as historical reference climate data for a baseline period (1996–2005). The data set encompasses temperature, precipitation, relative humidity, solar radiation, and wind speed. To ensure the accuracy of the baseline data set, the CFSR data was bias-corrected using climate records of three available local climate stations (Supplementary Section 1.1).

Climate projection datasets were taken from the Coordinated Regional Downscaling Experiment (CORDEX), which provides a suite of regional climate projections based on the Global Climate Models of the Coupled Model Intercomparison Project, Phase 5 (Taylor et al., 2012). We considered medium (RCP 4.5) and high (RCP 8.5) greenhouse gas emission scenarios from the IPCC - Fifth Assessment Record (AR5); and analysed the impacts in the near future (short; until 2030) and the mid future (medium; until 2050). The short-term time frame was selected to show the potential changes expected to occur in the coming next decade, and to show the necessity for immediate actions. The medium-term scenario was chosen to show the consequences of climate change at a time scale still relevant for today's population. Due to the capabilities of management and plants to adapt to changes in climate as well as long-term reactions of the farming community to adapt to new environmental conditions, we do not include a long-term impact assessment. For the short- and medium-term scenarios, we assumed that factors such as plant genetics and management strategies remain constant and at a current level.

The projections of future CO_2 concentration were based on (van Vuuren et al., 2011) and were assumed to be 420 ppm CO_2 (RCP 4.5) and 450 ppm CO_2 (RCP 8.5), during the time period 2021–2030, and 470 ppm CO_2 (RCP 4.5) and 520 ppm CO_2 (RCP 8.5) for the period 2041–2050. The study uses climate projections based on the older Representative Concentration Pathway (RCP) scenarios used in CMIP5 rather than the newer Shared Socio-economic Pathways (SSP) scenarios

from CMIP6. In terms of CO_2 emission scenarios, the scenarios RCP 4.5 and RCP 8.5 used in this study can be compared with the SSP2–4.5 and the early century SSP5–8.5 scenario, respectively (Fuglestvedt et al., 2021). We acknowledge that the newer climate projections might yield slightly different results, however, these are rather small given the mid-century time frame of the analysis (until 2050), in which CMIP5 and CMIP6 scenarios are still comparable (Fig. 3).

3. Results and discussion

3.1. Validation results of SWAT and APSIM

The skill of both the models to correctly reproduce plant water demands was validated against observed crop specific evapotranspiration (ET) rates, for the years 2009–2010. Validation results are given in terms of mean monthly absolute ET values and the coefficient of determination (\mathbb{R}^2) in Table 1. The skill of the models to correctly reproduce plant growths dynamics was validated against observed yield data, for the province Punjab, for the years 2002–2013. Validation results of yield simulations are given in terms of mean absolute yields and the relative error (RE) in Table 2 and are visualised in Fig. 2.

Both models show their strengths in their respective domains. SWAT performs slightly better in reproducing ET, due to its focus on hydrological processes and APSIM performs better in yields, due to its focus on plant growth dynamics. All models overestimate evapotranspiration rates and, with the exemption of the SWAT-Rice model, they all overestimate crop yields. This consistent overestimation of ET and yield is not surprising, considering the fact that we simulate hypothetical cases in which farm management practices are assumed to be close to optimal, with sufficient water and fertilizer supply. Even though this is not a perfect representation of reality, it is an important assumption in order to rule out yield variations due to management, while studying the potential risks of climate impacts on agricultural production. Note that in this study we use the models for studying relative changes for two future time periods, rather than studying the change in absolute values. The model structural reasons for the discrepancies in their performances are discussed in more detail further below. They give important insights into model uncertainties and their use as climate impact assessment tools.

3.2. Future climate trends in the LCC study area

Climate models project a strong increase in temperature over the



Fig. 3. Simulated yield estimates compared to observations; observations are the AGRIStats data = Agricultural Statistics of Pakistan. Bar heights show the mean of the years 2002-2013 and uncertainty bars show + /- one standard deviation.

Table 1

Validation results for simulated evapotranspiration rates.

Crop type	Observed ET	SWAT ET and $\ensuremath{\mathbb{R}}^2$		APSIM ET and R ²	
	Mean absolute ET [mm/month]	mean absolute ET [mm/ month]	R ²	Mean absolute ET [mm/ month]	R ²
Rice	75	83	0.65	84	0.26
Cotton	50	86	0.36	97	0.59
Maize	60	73	0.54	69	0.44

Table 2			
Validatio	n results for simula	nted yields.	
Cron	Observed Vield	SWAT Vield and RE	ADSIM VI

Crop type	Observed Yield	SWAT Yield and RE		APSIM Yield and RE	
	Mean absolute yield [kg/ha]	Mean absolute yield [kg/ha]	RE [%]	Mean absolute yield [kg/ha]	RE [%]
Rice	1813	1174	-35	1867	3
Cotton	666	950	30	867	23
Maize	3922	4092	4	6370	38

study region, under the high-emission scenario RCP 8.5 as well as under the moderate emission scenario RCP 4.5 (Fig. 1C). For the summer season (May-October), the ensemble means predict an increase of 1.0 °C (\pm 0.4 °C) for RCP 4.5 and 1.0 °C (\pm 0.3 °C) for RCP 8.5 until 2030, compared to the reference period of 1996-2005. A warming of 1.6 °C (\pm 0.5 °C) and 1.8 °C (\pm 0.5 °C) is projected for RCP 4.5 and 8.5, respectively, until 2050 (Fig. 4A). Strong increases in temperature under both scenarios point towards higher pressure on agricultural production resulting from increased heat stress on crop growth, especially during summer months (Ahmad et al., 2015; Awan et al., 2021; Yasin et al., 2022). During the four central months of the growing season (May-August), mean monthly temperatures are projected to exceed 32 °C (Fig. 4B). A high agreement between single climate model ensemble members regarding the consistent increase in future temperature, indicates that the future summer season warming in the LCC area can be projected with high confidence (Fig. 1C; Fig. 4A and B).

Precipitation projections, on the other hand, are highly uncertain and no clear trend in annual or monthly precipitation amounts can be detected (Fig. 1D, Fig. 4C and D). Estimations of future water availability based on precipitation projections for the study area, are therefore difficult. In this study, we assume that due to the constant irrigation activities in the LCC irrigation system, agricultural water availability is always assured, and plant water demand is met. Thus, impacts of changes in precipitation are small and water stress is kept low. Scenarios in which future water availability can no longer meet irrigation demands, e.g. due to shrinking surface or groundwater resources, are not analysed in this study. We focus exclusively on the effects of climate change on agricultural production given sufficient water supply and examine if the pressure on water resources will increase in future.

3.3. Declining yield levels under increasing temperatures

Both models (SWAT and APSIM) show that climate change will lead to a substantial reduction of future yield levels in the study area. Under current CO2 concentrations, mean yield levels of all three crops are projected to decrease by up to -24% (\pm 12%) under the high emission and mid-century scenario (Fig. 5A, light grey bar, RCP 8.5 2041-50). SWAT simulates yield reductions of up to -63% ($\pm 19\%$) for rice, -18% ($\pm 12\%$) for cotton, and -40% ($\pm 12\%$) for maize. APSIM simulates yield reductions of up to -22% ($\pm 4\%$) for rice, -20%(\pm 11%) for cotton, and - 12% (\pm 3%) for maize (Fig. 5 A-C). Despite differences in the predicted magnitudes of yield declines, SWAT and APSIM models agree in their trends (sign) and show that increasing yield losses align with increasing temperatures for all crop types. Considering that water demand and nutrient availability is assumed to be met, the results underline that increasing heat stress will play an important role in the decline of future crop yields. This is in-line with the findings of previous studies, which show that declining yields can be strongly related to temperature increases (Ortiz-Bobea et al., 2019; Saddique et al., 2020; Yasin et al., 2022; Zhao et al., 2017). Regional studies from Pakistan agree with our results and show that heat stress is the major factor threatening future yields in important agricultural regions in Pakistan (Awan et al., 2021; Khan et al., 2021; Yasin et al., 2022). Exact estimates of projected yield declines for our study region and our selected crops vary. The above cited studies suggest yield declines of approx. up to -17% to -25% in rice yields (Khan et al., 2021), up to 30% in cotton yields (Awan et al., 2021), and -8% to -55% in maize yields (Yasin et al., 2022), for mid-century climate change scenarios.

In addition it is worth noticing that the negative effects of rising temperatures on yields are often non-linear (Lobell et al., 2011; Schauberger et al., 2017; Schlenker and Roberts, 2009). Of particular importance is the exceedance of crop specific critical temperature thresholds, above which crops experience significant yield reductions. Cotton, which has a higher optimal temperature (approx. 30–32 °C, (Reddy et al., 1999; Schlenker and Roberts, 2009) and thus a higher heat resistance than rice (approx. 25 °C, Luo, 2011) and maize (approx. 29 °C, Schlenker and Roberts, 2009), can therefore better cope with elevated temperatures. This is also reflected in the results of this study,



Fig. 4. Projected temperature and precipitation change during Kharif (summer) months, for selected time periods 2021–2030 and 2041–2050, with respect to historical data (1996–2005). (A) Absolute seasonal temperature change and (B) absolute monthly temperature changes. (C) Relative seasonal precipitation changes and (D) absolute monthly precipitation changes. Red dots and displayed percentages show ensemble mean changes. Grey dots represent single ensemble members. The right panels show model ensemble uncertainty bands of 25th and 75th percentiles.

which show a strong crop specific sensitivity to heat stress, with stronger heat stress impacts for rice and maize and a lower heat stress impact for cotton. Furthermore, SWAT shows significantly higher sensitivities to temperature than APSIM, resulting in strong yield declines for all three crops. The stronger temperature sensitivity of SWAT explains the larger uncertainties in SWAT yield predictions, as temperature projections of the nine selected climate models are stronger reflected in the final SWAT results compared to ASPIM results. Despite the differences between APSIM and SWAT estimates, simulated yield losses of both models are unequivocal and would have large implications for future agricultural productivity in the study area.

3.4. Impact of rising CO_2 on plant development

Increasing atmospheric CO₂ levels dampen the negative effects of rising temperatures on yields and reveals the significant positive effect of elevated CO₂ concentrations on agricultural productivity (Fig. 5, dark grey bars). This effect is also known as CO2 fertilization. For the shortterm scenario (2021-2030), increasing CO2 concentrations could even prevent a significant decline in simulated crop yields. APSIM results show that average yield losses could be approx. 4.4% (RCP4.5, 2021-2030) to 6% (RCP8.5, 2021-2030) less severe due to increasing CO₂ concentrations. Averaging yield levels of all tree crops, SWAT projects a full offset of climate-related yield losses under enhanced CO2 levels. For the short-term scenario, SWAT simulates increasing yields of + 1.8% (RCP4.5) to + 5.5% (RCP8.5; Fig. 5A). Strong positive effects of increasing CO2 concentrations on plant growth, as seen in our results, have proven to counteract plant growth-limiting effects (Parry et al., 2004; Singh et al., 2020). At this point, it is worth to recall that, in this study, we assume a sufficient supply of nutrients at any time. Nutrient stress would limit the CO₂ fertilization effect (Reich et al., 2014), but in this study the positive effect of CO2 enrichment on plant development is

unimpeded. Overall, the effectiveness of CO_2 fertilization is still a large source of uncertainty (Elliott et al., 2014; McGrath and Lobell, 2013), which is yet to be improved in bio-physical crop models (Toreti et al., 2020). In the context of our study, uncertainty arises through the differences in representing CO_2 impacts on plant physiology by SWAT and APSIM.

SWAT accounts for the positive effects of CO_2 by increasing the radiation use efficiency of plants under rising CO_2 levels (Eqs. (S1) and (S2)). This increases the plants' ability to capture photosynthetically active radiation and leads to more efficient biomass production. Yet, the model equally applies this formulation to all crops, irrespective of their different photosynthetic pathways, i.e. C3- or C4-pathway. Hence, the agro-hydrological SWAT model does not account for plant-type-specific impacts of CO_2 and might overestimate the positive effects of CO_2 (Wu et al., 2012).

The APSIM crop models, on the other hand, consider plant-specific impacts (Vanuytrecht and Thorburn, 2017). In the case of C3-plants, such as rice and cotton, the models account for the photorespiratory part in the photosynthetic process (e.g. Eq. (S8) for cotton). Higher photorespiration lowers carbon assimilation under stress conditions and elevates it under higher CO₂ concentrations (Morison and Lawlor D.W, 1999). The APSIM maize-model (Fig. 5C) correctly assumes insensitivity to changing CO₂ effects for C4-plants. While this leads to unchanged maize yields with respect to CO₂ changes in APSIM simulations, SWAT assumes positive effects of CO₂ on maize growth. For cotton, rising CO₂ levels lead to increased cotton yields of approx. up to + 6% (APSIM) and + 14% (SWAT) until 2050, under the high emission scenario (Fig. 5B). Likewise, the simulated rice dynamics are consistent in both models. Rice yields (Fig. 5D) experience similar positive impacts under rising CO₂ levels compared to cotton (both C3-plants), but their yield reductions due to heat stress are too severe to be compensated by increasing CO₂ concentrations (Fig. 5D). Large discrepancies in the rice



Fig. 5. Projected changes in future crop yield under the RCP 4.5 and RCP 8.5 scenario, neglecting (light grey bars) and considering (dark grey bars) the impact of CO_2 changes. (A) Results are shown for all crops combined as well as separately for (B) cotton, (C) maize and (D) rice. Filled bars show the median estimate and black bars the respective 25th and 75th percentiles of the model ensemble (SWAT and APSIM with nine climate models). Separate results for SWAT and APSIM models are shown as coloured dots (median), and coloured line with the respective 25th and 75th percentiles for the model ensemble of nine climate models.

and maize simulations of the two models can be seen with respect to temperature changes. These are further discussed below.

Previous studies have indicated that increasing CO_2 improves water use efficiency by reducing plant transpiration which facilitates plant growth during drought conditions (Singh et al., 2020; Wullschleger et al., 2002; Yoo et al., 2009). At the same time, it has also been reported that reduced plant transpiration leads to increased heat stress, due to reduced evaporative cooling (Siebert et al., 2014; Vanuytrecht et al., 2012). These two effects are currently not captured by either of the models. As the strong increase in future temperature is projected under both RCP scenarios and together with the abundance of water due to irrigation, the positive effect of CO_2 on yield levels is most likely overestimated by both crop models.

Under further rising temperatures (scenarios for the period

2041–2050), APSIM and SWAT project a decline in crop yields indicating that even with further elevated CO₂ concentrations and unlimited water and nutrient availability, climate change induced yield declines cannot be prevented. APSIM projects an average yield decline of -8% ($\pm 3\%$, RCP 4.5) and -7% ($\pm 2.3\%$, RCP 8.5), while SWAT projects a similar yield decline with larger uncertainties of -7% ($\pm 13\%$, RCP 4.5) and -7% ($\pm 18\%$, RCP 8.5). All estimated crop yields show that the positive effects of increasing CO₂ levels are likely to diminish as temperatures continue to rise (Fig. 5A).

3.5. Future irrigation and evaporative demand

In the following, we analyse if rising temperatures will lead to an increase in irrigation demand and if irrigation is an effective measure to counteract the severe yield losses. Further, we discuss the reasons behind estimated yield declines in the context of changes in irrigation demand and evapotranspiration (ET).

Considering the significant temperature increase one would expect a strong increasing signal in plant water demand (Döll, 2002; Wada et al., 2013). Examining irrigation and evaporative demands in the study area, however, reveals that trends in future water demand do not align with projected temperature trends. Against the expectation of increasing irrigation needs due to rising temperatures, both crop models show that average irrigation demands increase less under higher temperatures. Under both emission scenarios and for both models, the strongest increase in irrigation demand is predicted for the moderate short-term scenario (RCP 4.5, 2021-2030), while less increase is predicted for the high-end emission scenario (RCP 8.5, 2041-2050). Fig. 6 displays the results for APSIM and SWAT separately to reveal important differences in their simulation results. Results are averaged over the selected summer crops cotton, maize and rice to focus on unraveling the complex dynamics of heat and water stress interactions under different climate scenarios. Crop specific results are presented in more detail the supplementary material (Figs. S4-S5).

For conditions with historical CO₂ levels, SWAT projects the lowest irrigation needs for the RCP 8.5 and mid-term future scenario (increase of irrigation needs by $1 \pm 8\%$; Fig. 6A). Under elevated CO₂ concentrations (Fig. 6B), water demand even reduces $(-4 \pm 7\%)$, which agrees with the effect of reduced plant water needs due to reduced stomatal conductance (Fernández-Martínez et al., 2019; Kimball et al., 2002). For the same scenario, irrigation demand simulated by APSIM will increase by approx. $5 \pm 5\%$. It appears insensitive to CO₂ changes, as irrigation demand remains constant regardless of changes in CO₂ levels (Fig. 6A vs. B). Yet, the significant increase in LAI under elevated CO₂ levels (Fig. 6B) as well as the negligible change in irrigation demand illustrates that APSIM likewise accounts for positive CO₂ effects on water demand and shows decreasing irrigation demand relative to leaf area growth. The reason for the surprisingly low increase in irrigation needs can be

explained by a similarly low increase in actual evapotranspiration rates (Fig. 6C and D). Irrespective of the strong temperature rise (up to 1.8 °C until 2041–2050), increases in ET are projected by both crop models to stay on average below 3% (\pm 4%; for model differences see Fig. 6), and do not increase with further rising temperatures.

Finally, the fact that biomass and yield strongly decline while future irrigation demands do not align with future temperature trends indicates that even if more water for intensified irrigation activity would be applied, it would not help to prevent yield losses when significant heat stress is the main yield controlling factor. This is further underlined by the strong impact of heat stress on leaf area growth and the already mentioned dynamics of biomass production, which are discussed in more detail below.

3.6. Future plant growth and plant productivity

SWAT model results suggest a significant reduction in LAI by up to -27% ($\pm 6\%$) under the high-emission scenario (median of all three crop models for RCP 8.5, 2041–2050). Crop specific declines are estimated to be up to -31% for maize, -25% for cotton, and -58% for rice (Fig. S8). The decreasing LAI trend clearly follows the increasing temperature trend, with the highest LAI reductions under the RCP 8.5 scenario (Fig. 6A), and reveals a strong sensitivity of leaf area growth to temperature changes (Fig. 6C). The reason for this strong temperature sensitivity is the following: SWAT estimates LAI development based on the influence of one predominant environmental stress factor (Eq. (S4)), which is heat stress in this study. Further, LAI calculations in SWAT do not account for CO₂ effects (Fig. 7A vs B), which results in strong LAI decreases even under higher CO₂ concentrations (Fig. 7B) and in decreasing ET rates (Fig. 7D).

APSIM on the other hand, which does not account for a specific heat stress factor in its LAI calculations, shows a clear LAI insensitivity to temperature (Fig. 7A vs. B). APSIM leaf area calculations are primarily controlled by water stress, but as constant water supply is guaranteed in



Fig. 6. (A and B) Projections of future irrigation demand and (C and D) future ET rates. (A and C) Changes under the baseline CO₂-scenario and (B and D) with increased CO₂-levels.



Fig. 7. (A and B) Projections of future LAI changes and (C and D) future biomass changes. (A and C) Changes under the baseline CO₂-scenario and (B and D) with increased CO₂-levels.

this study, LAI development is unimpeded. Yet, APSIM-LAI predictions show a notable sensitivity to CO₂-concentrations and increasing leaf growth under rising CO₂ levels, for C3-plants. Consequently, APSIM based simulated LAIs are projected to increase by up to 15% (\pm 10%) under the high emission scenario, which explains why APSIM-ET rates do not decrease despite ET reducing CO₂-effects (Fig. 5D). Crop specific results are -1% for maize (C4-plant), +24% for cotton, and +29% for rice (both C3-plants; Fig. S8). Similar effects were observed and described by Singh et al. (2020), revealing a strong increase in LAI due to CO₂ increases, offsetting higher water use efficiency.

As leaf area and temperature jointly control the evaporative demands and ultimately irrigation water needs, the differences in LAI simulations highlight the importance of model sensitivity to heat stress. The differences in the underlying model structures result in two dissimilar conclusions. On the one hand, SWAT projects a decline in future plant growth, which is strong enough to reduce ET and irrigation demand. This leads to the conclusion that irrigation intensification cannot help to mitigate future yield losses. APSIM, on the other hand, forecast increasing leaf area growth and indicates that intensified irrigation will be needed due to the resulting (albeit modest) rise in water demands. Yet, even in the latter case, intensified irrigation cannot prevent future yield losses.

The discrepancy between the projected increase in LAI and the decrease in biomass estimates by APSIM results from the way APSIM accounts for biomass partitioning processes. As a biophysical crop model, APSIM accounts for carbon assimilation in different plant fragments such as leaves, stem, fruits, and roots. The produced biomass can dynamically be allocated to different plant parts. Leaf area can therefore remain constant or even increase while the overall biomass decreases.

While SWAT and APSIM disagree in their LAI estimates, both models predict a substantial reduction in plant biomass under increasing temperatures (Fig. 7C). Under historical CO₂-levels SWAT predicts a reduction in biomass of up to -33% ($\pm 11\%$; up to -41% for maize,

-20% for cotton, and -64% for rice), while APSIM predicts a reduction of up to -25% ($\pm 11\%$; up to -17% for maize, -23% for cotton, and -5% for rice; RCP 8.5, 2041–2050). To this end, both models show a good agreement in their predicting trends and in their sensitivity towards CO₂ changes. Rising CO₂ levels might compensate negative temperature effects in the near future but for the mid-century scenario (2041-2050), biomass is projected to decline despite further elevated CO_2 levels (Fig. 7D). This aligns with observations which show a clear temperature dependent CO2 fertilization effect (Fernández-Martínez et al., 2019). As mentioned above, elevated CO₂ concentrations can lead to an enhanced carbon fixation rate under higher CO₂ concentrations. Yet, at the same time, it has been reported that reduced plant transpiration due to enhanced CO2 availability leads to increased heat stress. The reason is reduced evaporative cooling which induces higher leaf and plant organ temperatures (Siebert et al., 2014; Vanuytrecht et al., 2012). At the same time, the temperature increase can speed up the plant development and lead to less time for carbon assimilation and reduced biomass production. The results are supported by Morison and Lawlor D. W. (1999), who point out that a temperature increase of just a few tenths of a degree can act cumulatively over time and can significantly accelerate plant development and reduce time for biomass production.

Recalling that both models assume a sufficient supply of irrigation water and enough nutrient supply, our results reveal once more that heat stress is the dominating factor controlling future yield developments. We, therefore, conclude that the mitigation potential of irrigation to counteract future yield losses is severely limited under further rising temperatures. An intensification of irrigation might be able to mitigate yield declines in the near future, when positive CO₂ effects balance the harmful temperature effects and irrigation demands are still increasing (Fig. 6B). For the mid-century scenario, however, the positive CO₂ effects as well as irrigation intensification will not be able to compensate for the yield losses.

4. Conclusions

The study examines potential climate change impacts on cotton, maize and rice crops in the Punjab region of Pakistan. The main finding is that under expected climate change scenarios a substantial reduction in summer crop yields is likely to occur in the study region. It could be shown that plant development is dominantly controlled by heat stress and that negative climate change impacts on agricultural production can only be marginally mitigated by an intensification of irrigation. In the medium-term future, under further rising temperatures, irrigation cannot prevent increasing yield losses. Temperature related adaptation strategies such as the selection of more heat resistant crops, or changes in crop planting schedules seem therefore more suitable than water related adaptation measures. This aligns with current efforts in Pakistan to re-define eco-agrological zones according to changing climate patterns and to choose earlier planting dates to avoid summer heat waves.

By using two models from different scientific disciplines, and with two different model architectures, the study is able to point out important model structural limitations which influence the final results of climate impacts on local crop production and water demands. LAI simulations and the simulation of CO2-sensitivities were identified as important model routines, leading to discrepancies in the results of our climate change impact assessment using the SWAT and the APSIM models. This also leads to important differences in the estimation of irrigation water demand. We therefore conclude by suggesting a better combination of the strengths of hydrological and biophysical crop models. The integration of more robust hydrological routines into biophysical crop models is a promising way forward. Such a hard-coupled model integration will be valuable for agricultural climate change impact studies as it enables a simultaneous consideration of the hydroclimatic environment with specific bio-physical responses of plants to climate change.

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

The data is freely available and I have shared the link to the repository in the manuscript.

Acknowledgments

The research was conducted in the scope of the joint research project "Innovative Impulses Reducing the Water Footprint of the Global Cotton-Textile Industry towards the UN Sustainable Development Goals (InoCottonGROW)" and is supported by the German Federal Ministry of Education and Research (BMBF) under the terms of its funding initiative "Global Resource Water (GROW)" [BMBF reference no. 02WGR1422]. We thank the entire InoCottonGROW team and especially the partners from the University of Agriculture Faisalabad (UAF) for their friendly collaboration and support. We furthermore thank two anonymous reviewers for their valuable feedbacks, which helped to improve the manuscript. Important advice with respect to real-world plant physiological processes, was kindly given by Bernhard Schauberger, Diego Miralles and Akash Koppa. We acknowledge the World Climate Research Program's Working Group on Regional Climate, and the Working Group on Coupled Modelling. We also thank the climate modelling groups, (listed in the legend of Fig. S2 of this paper) for producing and making their model output available. Acknowledgment is furthermore made to the APSIM Initiative and the SWAT model development team. Final gratitude to Maren Schneider for the illustration of Fig. 2.

Open Research

The data produced in this study can be accessed through the following data repository: https://doi.org/10.5281/zenodo.4603 703.

Appendix A. Supporting information

Supplementary data associated with this article can be found in the online version at doi:10.1016/j.agwat.2023.108243.

References

- Ahmad, A., Ashfaq, M., Rasul, G., Wajid, S.A., Khaliq, T., Rasul, F., Saeed, U., Rahman, M.Hu, Hussain, J., Ahmad Baig, I., Naqvi, S.A.A., Bokhari, S.A.A., Ahmad, S., Naseem, W., Hoogenboom, G., Valdivia, R.O., 2015. Impact of Climate Change on the Rice–Wheat Cropping System of Pakistan, Series on Climate Change Impacts, Adaptation, and Mitigation. IMPERIAL COLLEGE PRESS., pp. 219–258.
- Arnold, J.G., Moriasi, D.N., Gassman, P.W., Abbaspour, K.C., White, M.J., Srinivasan, R., Santhi, C., Harmel, R.D., van Griensven, A., Van Liew, M.W., Kannan, N., Jha, M.K., 2012. SWAT: model use, calibration, and validation. Trans. ASABE 55 (4), 1491–1508. https://doi.org/10.13031/2013.42256.
- Awan, Z., Khaliq, T., Masood Akhtar, M., Imran, A., Irfan, M., Jarrar Ahmed, M., Ahmad, A., 2021. Building climate-resilient cotton production system for changing climate scenarios using the DSSAT model. Sustainability 13 (19), 10495. https://doi. org/10.3390/su131910495.
- Becker, R., Koppa, A., Schulz, S., Usman, M., Beek, T. aus der, Schüth, C., 2019. Spatially distributed model calibration of a highly managed hydrological system using remote sensing-derived ET data. J. Hydrol. 577, 123944 https://doi.org/10.1016/j. ihydrol.2019.123944.
- Boote, K.J., Adam, M., Ahmad, I., Ahmad, S., Cammarano, D., Chattha, A.A., Claessens, L., Dimes, J., Durand, W., Freduah, B.S., Gummadi, S., Hargreaves, J., Hoogenboom, G., Homann-Kee Tui, S., Jones, J.W., Khaliq, T., MacCarthy, D.S., Masikati, P., McDermid, S., Kadiyala, D.M., Nenkam, A., Porter, C., Ruane, A.C., Subash, N., Thorburn, P., Traore, P.S., Vellingiri, G., Wajid, S.A., 2021. Understanding Differences in Climate Sensitivity Simulations of APSIM and DSSAT Crop Models. WORLD SCIENTIFIC.
- Carter, E.K., Melkonian, J., Riha, S.J., Shaw, S.B., 2016. Separating heat stress from moisture stress: analyzing yield response to high temperature in irrigated maize. Environ. Res. Lett. 11 (9), 94012. https://doi.org/10.1088/1748-9326/11/9/ 094012.
- Deihimfard, R., Eyni-Nargeseh, H., Mokhtassi-Bidgoli, A., 2018. Effect of future climate change on wheat yield and water use efficiency under semi-arid conditions as predicted by APSIM-wheat model. Int. J. Plant Prod. 12 (2), 115–125. https://doi. org/10.1007/s42106-018-0012-4.
- Döll, P., 2002. Impact of climate change and variability on irrigation requirements: a global perspective. Clim. Change 54 (3), 269–293. https://doi.org/10.1023/A: 1016124032231.
- Elliott, J., Deryng, D., Müller, C., Frieler, K., Konzmann, M., Gerten, D., Glotter, M., Flörke, M., Wada, Y., Best, N., Eisner, S., Fekete, B.M., Folberth, C., Foster, I., Gosling, S.N., Haddeland, I., Khabarov, N., Ludwig, F., Masaki, Y., Olin, S., Rosenzweig, C., Ruane, A.C., Satoh, Y., Schmid, E., Stacke, T., Tang, Q., Wisser, D., 2014. Constraints and potentials of future irrigation water availability on agricultural production under climate change. Proc. Natl. Acad. Sci. 111 (9), 3239–3244. https://doi.org/10.1073/pnas.1222474110.
- Fader, M., Shi, S., Bloh, W., von, Bondeau, A., Cramer, W., 2016. Mediterranean irrigation under climate change: more efficient irrigation needed to compensate for increases in irrigation water requirements. Hydrol. Earth Syst. Sci. 20 (2), 953–973. https://doi.org/10.5194/hess-20-953-2016.
- Faria, R.T., Bowen, W.T., 2003. Evaluation of DSSAT soil-water balance module under cropped and bare soil conditions. Braz. Arch. Biol. Technol. 46 (4), 489–498. https://doi.org/10.1590/S1516-89132003000400001.
- Fernández-Martínez, M., Sardans, J., Chevallier, F., Ciais, P., Obersteiner, M., Vicca, S., Canadell, J.G., Bastos, A., Friedlingstein, P., Sitch, S., Piao, S.L., Janssens, I.A., Peñuelas, J., 2019. Global trends in carbon sinks and their relationships with CO₂ and temperature. Nat. Clim. Change 9 (1), 73–79. https://doi.org/10.1038/s41558-018-0367-7.
- Fischer, G., Tubiello, F.N., van Velthuizen, H., Wiberg, D.A., 2007. Climate change impacts on irrigation water requirements: Effects of mitigation, 1990–2080. Technol. Forecast. Soc. Change 74 (7), 1083–1107. https://doi.org/10.1016/j. techfore.2006.05.021.
- Fuglestvedt et al., 2021, IPCC, Chapter 1: Framing and Context Special Report on Climate Change and Land: Cross-Chapter Box 1.4, https://www.ipcc.ch/srccl/ chapter/chapter-1/. Accessed 2/10/2023.
- Gassman, P.W., Sadeghi, A.M., Srinivasan, R., 2014. Applications of the SWAT model special section: overview and insights. J. Environ. Qual. 8.
- Gaydon, D.S., Khaliq, T., Ahmad, M.-D., Cheema, M., Gull, U., 2021. Tweaking Pakistani Punjab rice-wheat management to maximize productivity within nitrate leaching limits. Field Crops Res. 260, 107964 https://doi.org/10.1016/j.fcr.2020.107964.
- Hanjra, M.A., Qureshi, M.E., 2010. Global water crisis and future food security in an era of climate change. Food Policy 35 (5), 365–377. https://doi.org/10.1016/j. foodpol.2010.05.006.

- Holzworth, D.P., Huth, N.I., deVoil, P.G., Zurcher, E.J., Herrmann, N.I., McLean, G., Chenu, K., van Oosterom, E.J., Snow, V., Murphy, C., Moore, A.D., Brown, H., Whish, J.P., Verrall, S., Fainges, J., Bell, L.W., Peake, A.S., Poulton, P.L., Hochman, Z., Thorburn, P.J., Gaydon, D.S., Dalgliesh, N.P., Rodriguez, D., Cox, H., Chapman, S., Doherty, A., Teixeira, E., Sharp, J., Cichota, R., Vogeler, I., Li, F.Y., Wang, E., Hammer, G.L., Robertson, M.J., Dimes, J.P., Whitbread, A.M., Hunt, J., van Rees, H., McClelland, T., Carberry, P.S., Hargreaves, J.N., MacLeod, N., McDonald, C., Harsdorf, J., Wedgwood, S., Keating, B.A., 2014. APSIM – Evolution towards a new generation of agricultural systems simulation. Environ. Model. Softw. 62, 327–350. https://doi.org/10.1016/j.envsoft.2014.07.009.
- Immerzeel, W.W., van Beek, L.P.H., Bierkens, M.F.P., 2010. Climate change will affect the asian water towers. Science 328 (5984), 1382–1385. https://doi.org/10.1126/ science.1183188.
- Khaliq, T., Gaydon, D.S., Ahmad, M.-D., Cheema, M., Gull, U., 2019. Analyzing crop yield gaps and their causes using cropping systems modelling–A case study of the Punjab rice-wheat system, Pakistan. Field Crops Res. 232, 119–130. https://doi.org/ 10.1016/j.fcr.2018.12.010.
- Khan, M.A., Khan, J.A., Ali, Z., Ahmad, I., Ahmad, M.N., 2016. The challenge of climate change and policy response in Pakistan. Environ. Earth Sci. 75 (5), 412. https://doi. org/10.1007/s12665-015-5127-7.
- Khan, N.A., Gao, Q., Abid, M., Shah, A.A., 2021. Mapping farmers' vulnerability to climate change and its induced hazards: evidence from the rice-growing zones of Punjab, Pakistan. Environ. Sci. Pollut. Res 28 (4), 4229–4244. https://doi.org/ 10.1007/s11356-020-10758-4.
- Kimball, B.A., Kobayashi, K., Bindi, M., 2002. Responses of agricultural crops to free air CO₂ enrichment. Adv. Agron. 77, 293–361.
- Lesk, C., Anderson, W., Rigden, A., Coast, O., Jägermeyr, J., McDermid, S., Davis, K.F., Konar, M., 2022. Compound heat and moisture extreme impacts on global crop yields under climate change. Nat Rev Earth Environ, 3 (12), 872–889. https://doi. org/10.1038/s43017-022-00368-8.
- Lobell, D.B., Bänziger, M., Magorokosho, C., Vivek, B., 2011. Nonlinear heat effects on African maize as evidenced by historical yield trials. Nat. Clim. Change 1 (1), 42–45. https://doi.org/10.1038/nclimate1043.
- McGrath, J.M., Lobell, D.B., 2013. Regional disparities in the CO₂ fertilization effect and implications for crop yields. Environ. Res. Lett. 8 (1), 14054. https://doi.org/ 10.1088/1748-9326/8/1/014054.
- Mekonnen, M.M., Hoekstra, A.Y., 2011. The green, blue and grey water footprint of crops and derived crop products. Hydrol. Earth Syst. Sci. 15 (5), 1577–1600. https://doi. org/10.5194/hess-15-1577-2011.
- MNFSR, 2021, Ministry of National Food Security and Research., http://www.mnfsr.gov. pk/pubDetails.aspx. Accessed 12/22/2020.
- Molden, D., Oweis, T., Steduto, P., Bindraban, P., Hanjra, M.A., Kijne, J., 2010. Improving agricultural water productivity: Between optimism and caution. Agric. Water Manag. 97 (4), 528–535. https://doi.org/10.1016/j.agwat.2009.03.023.
- Morison, J., Lawlor, D.W., 1999. Interactions between increasing CO 2 concentration and temperature on plant growth. Plant. Cell Environ. 22 (6), 659–682. https://doi.org/ 10.1046/j.1365-3040.1999.00443.x.
- Neitsch, S.L., Arnold, J.G., Kiniry, J.R., Williams, J.R., 2009. Soil and water assessment tool. theoretical documentation. Version 2009. Tex. Water Resour. Insitute Tec. Rep. 406 (No.).
- Ortiz-Bobea, A., Wang, H., Carrillo, C.M., Ault, T.R., 2019. Unpacking the climatic drivers of US agricultural yields. Environ. Res. Lett. 14 (6), 64003. https://doi.org/ 10.1088/1748-9326/ab1e75.
- Oxfam, 2009. Climate change, poverty and environmental crisis in the disaster prone areas of Pakistan. Pak. Program. Off., Islamabad.
- Parry, M., Rosenzweig, C., Iglesias, A., Livermore, M., Fischer, G., 2004. Effects of climate change on global food production under SRES emissions and socio-economic scenarios. Glob. Environ. Change 14 (1), 53–67. https://doi.org/10.1016/j. gloenycha.2003.10.008.
- Qureshi, A.S., 2011. Water management in the indus basin in pakistan: challenges and opportunities. Mt. Res. Dev. 31 (3), 252–260. https://doi.org/10.1659/MRD-JOURNAL-D-11-00019.1.
- Reddy, K.R., Davidonis, G.H., Johnson, A.S., Vinyard, B.T., 1999. Temperature regime and carbon dioxide enrichment alter cotton boll development and fiber properties. Agron. J. 91 (5), 851–858. https://doi.org/10.2134/agronj1999.915851x.
- Reich, P.B., Hobbie, S.E., Lee, T.D., 2014. Plant growth enhancement by elevated CO₂ eliminated by joint water and nitrogen limitation. Nat. Geosci. 7 (12), 920–924. https://doi.org/10.1038/ngeo2284.
- Rosenzweig, C., Elliott, J., Deryng, D., Ruane, A.C., Müller, C., Arneth, A., Boote, K.J., Folberth, C., Glotter, M., Khabarov, N., Neumann, K., Piontek, F., Pugh, T.A.M., Schmid, E., Stehfest, E., Yang, H., Jones, J.W., 2014. Assessing agricultural risks of climate change in the 21st century in a global gridded crop model intercomparison. Proc. Natl. Acad. Sci. 111 (9), 3268–3273. https://doi.org/10.1073/ pnas.1222463110.
- Saddique, Q., Khan, M.I., Habib ur, Rahman, Jiatun, M., Waseem, X., Gaiser, M., Mohsin Waqas, T., Ahmad, M., Chong, I., Cai, H. L., 2020. Effects of Elevated Air Temperature and CO₂ on Maize Production and Water Use Efficiency under Future Climate Change Scenarios in Shaanxi Province. China Atmosphere 11 (8), 843. https://doi.org/10.3390/atmos11080843.
- Saeed, F., Athar, H., 2018. Assessment of simulated and projected climate change in Pakistan using IPCC AR4-based AOGCMs. Theor. Appl. Climatol. 134 (3–4), 967–980. https://doi.org/10.1007/s00704-017-2320-5.
- Saha, S., Moorthi, S., Pan, H.-L., Wu, X., Wang, J., Nadiga, S., Tripp, P., Kistler, R., Woollen, J., Behringer, D., Liu, H., Stokes, D., Grumbine, R., Gayno, G., Wang, J., Hou, Y.-T., Chuang, H., Juang, H.-M.H., Sela, J., Iredell, M., Treadon, R., Kleist, D., van Delst, P., Keyser, D., Derber, J., Ek, M., Meng, J., Wei, H., Yang, R., Lord, S., van

den Dool, H., Kumar, A., Wang, W., Long, C., Chelliah, M., Xue, Y., Huang, B., Schemm, J.-K., Ebisuzaki, W., Lin, R., Xie, P., Chen, M., Zhou, S., Higgins, W., Zou, C.-Z., Liu, Q., Chen, Y., Han, Y., Cucurull, L., Reynolds, R.W., Rutledge, G., Goldberg, M., 2010. The NCEP Climate Forecast System Reanalysis. Bull. Am. Meteorol. Soc. 91 (8), 1015–1058. https://doi.org/10.1175/2010BAMS3001.1.

- Schauberger, B., Archontoulis, S., Arneth, A., Balkovic, J., Ciais, P., Deryng, D., Elliott, J., Folberth, C., Khabarov, N., Müller, C., Pugh, T.A.M., Rolinski, S., Schaphoff, S., Schmid, E., Wang, X., Schlenker, W., Frieler, K., 2017. Consistent negative response of US crops to high temperatures in observations and crop models. Nat. Commun. 8 (1), 13931. https://doi.org/10.1038/ncomms13931.
- Schewe, J., Heinke, J., Gerten, D., Haddeland, I., Arnell, N.W., Clark, D.B., Dankers, R., Eisner, S., Fekete, B.M., Colón-González, F.J., Gosling, S.N., Kim, H., Liu, X., Masaki, Y., Portmann, F.T., Satoh, Y., Stacke, T., Tang, Q., Wada, Y., Wisser, D., Albrecht, T., Frieler, K., Piontek, F., Warszawski, L., Kabat, P., 2014. Multimodel assessment of water scarcity under climate change. Proc. Natl. Acad. Sci. 111 (9), 3245–3250. https://doi.org/10.1073/pnas.1222460110.
- Schlenker, W., Roberts, M.J., 2009. Nonlinear temperature effects indicate severe damages to U.S. crop yields under climate change. Proc. Natl. Acad. Sci. 106 (37), 15594–15598. https://doi.org/10.1073/pnas.0906865106.
- Schulz, S., Becker, R., Richard-Cerda, J.C., Usman, M., Beek, T., Merz, R., Schüth, C., 2021. Estimating water balance components in irrigated agriculture using a combined approach of soil moisture and energy balance monitoring, and numerical modeling. Hydrol. Process. https://doi.org/10.1002/hyp.14077.
- Shukla, P.R., Skea, J., Calvo Buendia, E., Masson-Delmotte, V., Pörtner, H.-O., Roberts, D.C., Zhai, P., Slade, R., Connors, S., van Diemen, R., Ferrat, M., Haughey, E., Luz, S., Neogi, S., Pathak, M., Petzold, J., Portugal Pereira, J., Vyas, P., Huntley, E., Kissick, K., Belkacemi, M., Malley, J. 2019. Climate Change and Land: an IPCC special report on climate change, desertification, land degradation, sustainable land management, food security, and greenhouse gas fluxes in terrestrial ecosystems.
- Siebert, S., Ewert, F., Eyshi Rezaei, E., Kage, H., Graß, R., 2014. Impact of heat stress on crop yield—on the importance of considering canopy temperature. Environ. Res. Lett. 9 (4), 44012. https://doi.org/10.1088/1748-9326/9/4/044012.
- Singh, A., Kumar, S., Akula, S., Lawrence, D.M., Lombardozzi, D.L., 2020. Plant growth nullifies the effect of increased water-use efficiency on streamflow under elevated CO₂ in the Southeastern United States. Geophys. Res. Lett. 10.
- Tack, J., Barkley, A., Hendricks, N., 2017. Irrigation offsets wheat yield reductions from warming temperatures. Environ. Res. Lett. 12 (11), 114027 https://doi.org/ 10.1088/1748-9326/aa8d27.
- Taraz, V., 2018. Can farmers adapt to higher temperatures? Evidence from India. World Dev. 112, 205–219. https://doi.org/10.1016/j.worlddev.2018.08.006.
- Taylor, K.E., Stouffer, R.J., Meehl, G.A., 2012. An Overview of CMIP5 and the Experiment Design. Bull. Am. Meteorol. Soc. 93 (4), 485–498. https://doi.org/ 10.1175/BAMS-D-11-00094.1.
- Toreti, A., Deryng, D., Tubiello, F.N., Müller, C., Kimball, B.A., Moser, G., Boote, K., Asseng, S., Pugh, T.A.M., Vanuytrecht, E., Pleijel, H., Webber, H., Durand, J.-L., Dentener, F., Ceglar, A., Wang, X., Badeck, F., Lecerf, R., Wall, G.W., van den Berg, M., Hoegy, P., Lopez-Lozano, R., Zampieri, M., Galmarini, S., O'Leary, G.J., Manderscheid, R., Mencos Contreras, E., Rosenzweig, C., 2020. Narrowing uncertainties in the effects of elevated CO₂ on crops. Nat. Food 1 (12), 775–782. https://doi.org/10.1038/s43016-020-00195-4.
- van Vuuren, D.P., Edmonds, J., Kainuma, M., Riahi, K., Thomson, A., Hibbard, K., Hurtt, G.C., Kram, T., Krey, V., Lamarque, J.-F., Masui, T., Meinshausen, M., Nakicenovic, N., Smith, S.J., Rose, S.K., 2011. The representative concentration pathways: an overview. Clim. Change 109 (1–2), 5–31. https://doi.org/10.1007/ s10584-011-0148-z.
- Vanuytrecht, E., Thorburn, P.J., 2017. Responses to atmospheric CO_2 concentrations in crop simulation models: a review of current simple and semicomplex representations and options for model development. Glob. Change Biol. 23 (5), 1806–1820. https:// doi.org/10.1111/gcb.13600.
- Vanuytrecht, E., Raes, D., Willems, P., Geerts, S., 2012. Quantifying field-scale effects of elevated carbon dioxide concentration on crops. Clim. Res. 54 (1), 35–47. https:// doi.org/10.3354/cr01096.
- Wada, Y., Wisser, D., Eisner, S., Flörke, M., Gerten, D., Haddeland, I., Hanasaki, N., Masaki, Y., Portmann, F.T., Stacke, T., Tessler, Z., Schewe, J., 2013. Multimodel projections and uncertainties of irrigation water demand under climate change: Irrigation demand under climate change. Geophys. Res. Lett. 40 (17), 4626–4632. https://doi.org/10.1002/grl.50686.
- Wang, X., Zhang, J., Ali, M., Shahid, S., He, R., Xia, X., Jiang, Z., 2016. Impact of climate change on regional irrigation water demand in Baojixia irrigation district of China. Mitig. Adapt. Strateg. Glob. Change 21 (2), 233–247. https://doi.org/10.1007/ s11027-014-9594-z.
- Wu, Y., Liu, S., Abdul-Aziz, O.I., 2012. Hydrological effects of the increased CO₂ and climate change in the Upper Mississippi River Basin using a modified SWAT. Clim. Change 110 (3–4), 977–1003. https://doi.org/10.1007/s10584-011-0087-8.
 Wullschleger, S.D., Tschaplinski, T.J., Norby, R.J., 2002. Plant water relations at
- Wullschleger, S.D., Tschaplinski, T.J., Norby, R.J., 2002. Plant water relations at elevated CO 2 - implications for water-limited environments: Interactions between elevated CO 2 and drought. Plant, Cell Environ. 25 (2), 319–331. https://doi.org/ 10.1046/j.1365-3040.2002.00796.x.
- Yasin, M., Ahmad, A., Khaliq, T., Habib-ur-Rahman, M., Niaz, S., Gaiser, T., Ghafoor, I., Hassan, H.Su, Qasim, M., Hoogenboom, G., 2022. Climate change impact uncertainty assessment and adaptations for sustainable maize production using multi-crop and climate models. Environ. Sci. Pollut. Res 29 (13), 18967–18988. https://doi.org/10.1007/s11356-021-17050-z.

R. Becker et al.

- Yoo, C.Y., Pence, H.E., Hasegawa, P.M., Mickelbart, M.V., 2009. Regulation of transpiration to improve crop water use. Crit. Rev. Plant Sci. 28 (6), 410–431. https://doi.org/10.1080/07352680903173175.
- Zaveri, E., Lobell, D., 2019. The role of irrigation in changing wheat yields and heat sensitivity in India. Nature. Communications 10 (1), 9. https://doi.org/10.1038/ s41467-019-12183-.
- Zhao, C., Liu, B., Piao, S., Wang, X., Lobell, D.B., Huang, Y., Huang, M., Yao, Y., Bassu, S., Ciais, P., Durand, J.-L., Elliott, J., Ewert, F., Janssens, I.A., Li, T., Lin, E., Liu, Q., Martre, P., Müller, C., Peng, S., Peñuelas, J., Ruane, A.C., Wallach, D., Wang, T., Wu, D., Liu, Z., Zhu, Y., Zhu, Z., Asseng, S., 2017. Temperature increase reduces global yields of major crops in four independent estimates. Proc. Natl. Acad. Sci. 114 (35), 9326–9331. https://doi.org/10.1073/pnas.1701762114.